Author’s response to reviews

Title: A randomized controlled trial to examine the effect of two teaching methods on preschool children’s language and communication, executive functions, socioemotional comprehension, and early math skills

Authors:

Tove Gerholm (tove@ling.su.se)

Petter Kallioinen (petter@ling.su.se)

Signe Tonér (signe.toner@ling.su.se)

Sofia Frankenberg (sofia.frankenberg@buv.su.se)

Susanne Kjällander (susanne.kjallander@buv.su.se)

Anna Palmer (anna.palmer@buv.su.se)

Hillevi Lenz-Taguchi (hillevi.lenz-taguchi@buv.su.se)

Version: 3 Date: 13 Jul 2019

Author’s response to reviews:

Review comments:

1. Please ensure that all relevant endnotes are included for publication, at the moment it seems that placeholders have been included for table 2.

Reply: I believe I have fixed this now but I'm not completely sure I understood the question. I had written the placeholder-thing since I couldn't figure out how to make endnotes NOT follow each other (as ix and x belong to a table that is lifted to the end of the document although they should be placed early in the paper, if you see what I mean...). Hopefully some clever person on your end can make this work in the actual manuscript - if what I did now isn't sufficient.

2. When submitting your revised manuscript please ensure you do so as a single clean copy without any tracked changes, colored or highlighted text, as these are no longer required at this stage of the editorial process.

Reply: This is now done.